RESEARCH EDUCATION TREATMENT

ADVOCACY



## Assessing Clinically Meaningful Treatment Effects in Controlled Trials: Chronic Migraine as an Example

David W. Dodick,\* Catherine C. Turkel,<sup>†</sup> Ronald E. DeGryse,<sup>†</sup> Hans-Christoph Diener,<sup>‡</sup> Richard B. Lipton,<sup>§</sup> Sheena K. Aurora,<sup>¶</sup> Marissa E. Nolan,<sup>||</sup> and Stephen D. Silberstein\*\*

\*Department of Neurology, Mayo Clinic Arizona, Phoenix, Arizona.

<sup>†</sup>Allergan, Inc, Irvine, California.

American

Pain 🤇

Society

<sup>‡</sup>Department of Neurology and Stroke Center, University of Essen, Essen, Germany.

<sup>§</sup>Department of Neurology and the Montefiore Headache Center, Albert Einstein College of Medicine, Bronx, New York.

<sup>¶</sup>Department of Neurology and Neurological Sciences, Stanford University, Stanford, California.

<sup>®</sup>Formerly of the Department of Medical Affairs, Imprint Publication Science, New York, New York.

\*\*Department of Neurology, Thomas Jefferson University, Philadelphia, Pennsylvania.

Abstract: In addition to headache, persons with chronic migraine (CM) experience multiple symptoms, both ictal and interictal, that may contribute to their suffering. Translating clinical trial results into practice requires assessment of the results' clinical meaningfulness. When examining treatment benefit in this disabled patient population, multiple headache-symptom measures should be considered to fully reflect clinical relevance. Currently, only onabotulinumtoxinA is approved specifically for headache prophylaxis in adults with CM. Topiramate is the only other therapeutic agent with double-blind, placebo-controlled evidence in this population. Herein we evaluate the clinical meaningfulness of onabotulinumtoxinA and topiramate as headache prophylaxis in CM by comparing primary endpoints from the placebo-controlled, double-blind phase of the Phase 3 REsearch Evaluating Migraine Prophylaxis Therapy (PREEMPT) clinical program and the topiramate clinical trial (frequency of headache days [primary endpoint in PREEMPT; secondary in topiramate trial] and migraine/ migrainous days [primary in topiramate trial, or "migraine/probable-migraine days"; secondary in

Received February 25, 2014; Revised September 2, 2014; Accepted November 5, 2014.

R.B.L. receives research support from the NIH [PO1 AG03949 (Program Director), RO1AG025119 (Investigator), RO1AG022374-06A2 (Investigator), RO1AG034119 (Investigator), RO1AG12101 (Investigator), K23AG030857 (Mentor), K23NS05140901A1 (Mentor), and K23NS47256 (Mentor), the National Headache Foundation, and the Migraine Research Fund; serves on the editorial boards of *Neurology* and *Cephalalgia* and as senior advisor to *Headache*; has reviewed for the NIA and NINDS; holds stock options in eNeura Therapeutics (a company without commercial products); and serves as consultant, advisory board member, or has received honoraria from Allergan, American Headache Society, Autonomic Technologies, Boehringer Ingelheim, Boston Scientific, Bristol Myers Squibb, Cognimed, Colucid, Eli Lilly, Endo, eNeura Therapeutics, GlaxoSmithKline, MAP, Merck, Nautilus Neuroscience, Novartis, NuPathe, Pfizer, and Vedanta.

S.K.A. has received research support from Advanced Bionics, Allergan, Inc, Boston Scientific, eNeura, GlaxoSmithKline, MAP Pharmaceuticals, Inc, Merck, NuPathe, Inc, and the NIH; has served as a consultant for Allergan, Arteaus Therapeutics, Merck, and eNeura; and has received honoraria from Merck, GlaxoSmithKline, Allergan, Inc, Nautilus, and Zogenix, Inc. M.E.N. was an employee of Imprint Publication Science at the time of this research. Her work was funded by Allergan.

S.D.S. is on the advisory panel of and receives honoraria from Allergan, Amgen, Capnia, Coherex, GlaxoSmithKline, Iroko Pharmaceuticals, Lilly, MAP, Medtronic, Merck, Neuralieve, NINDS, NuPathe, Pfizer, and St. Jude Medical. He serves as a consultant for and receives honoraria from Amgen, MAP, Nautilus, Novartis, Opti-Nose, and Zogenix. Dr. Silberstein's employer receives research support from Allergan, BMS, Cumberland, ElectroCore, Lilly, Merck, Opti-Nose, St. Jude Medical, and Troy Healthcare.

Address reprint requests to David W. Dodick, MD, Mayo Clinic, Arizona— Neurology, Department of Neurology, 5777 E. Mayo Boulevard, Phoenix, AZ 85054. E-mail: dodick.david@mayo.edu

1526-5900/\$36.00

© 2015 by the American Pain Society http://dx.doi.org/10.1016/j.jpain.2014.11.004

Funded by Allergan, Inc.

D.W.D., within the past 12 months, has served on advisory boards and has consulted for Allergan, Amgen, Alder, Arteaus, Pfizer, Merck, ENeura, NuPathe, Eli Lilly & Company, Autonomic Technologies, WL Gore, Ethicon J&J, Zogenix, Supernus, Labrys, and Bristol-Myers Squibb. D.W.D. has received funding for travel, speaking, or editorial activities from the following: CogniMed, Scientiae, IntraMed, SAGE Publishing, Sun Pharma, Allergan, Lippincott Williams & Wilkins, Oxford University Press, Cambridge University Press, Miller Medical, and Annenberg for Health Sciences; he serves as Editor-in-Chief and on the editorial boards of *The Neurologist, Lancet Neurology*, and *Postgraduate Medicine* and has served as Editor-in-Chief of *Headache Currents* and as an Associate Editor of *Headache*. He receives publishing royalties for *Wolff's Headache*, 8th (Cambridge University Press, 2009) and *Handbook of Headache* (Cambridge University Press, 2010). C.C.T. is an employee of, holds stock/ stock options in, and holds a patent with Allergan, Inc.

R.E.D. is an employee of and holds stock/stock options in Allergan, Inc.

H.C.D. received honoraria for participation in clinical trials, contribution to advisory boards, or oral presentations from Addex Pharma, Allergan, Almirall, Autonomic Technology, AstraZeneca, Bayer Vital, Berlin Chemie, Boehringer Ingelheim, Bristol-Myers Squibb, Coherex, CoLucid, GlaxoSmithKline, Grünenthal, Janssen-Cilag, Lilly, La Roche, 3M Medica, Medtronic, Menerini, Minster, MSD, Neuroscore, Novartis, Johnson & Johnson, Pierre Fabre, Pfizer, Schaper and Brümmer, Sanofi, St. Jude, and Weber & Weber. He has received financial support for research projects from Allergan, Almirall, AstraZeneca, Bayer, GSK, Janssen-Cilag, MSD, and Pfizer. Headache research at the Department of Neurology in Essen is supported by the German Research Council (DFG), the German Ministry of Education and Research (BMBF), and the European Union. H.C.D. has no ownership interest and does not own stocks of any pharmaceutical company.

#### Dodick et al

PREEMPT]). Additionally, outcome measures such as responder rates, health-related quality of life, discontinuation rates, safety, and tolerability profiles are important clinical considerations. The clinical data indicate that statistically significant, clinically relevant treatment benefits exist for both onabotu-linumtoxinA and topiramate. These data support these treatments as meaningful headache prophylaxis in adults with CM.

**Perspective:** CM is a chronic pain condition. We sought to determine the clinical relevance of recent trials in this disabled population. Clinical data indicate that statistically significant, clinically relevant treatment benefits exist for both onabotulinumtoxinA and topiramate, and support use of these treatments as meaningful headache prophylaxis in CM.

#### © 2015 by the American Pain Society

Key words: Chronic migraine, onabotulinumtoxinA, topiramate, prophylactic treatment, clinical meaningfulness.

hronic migraine (CM) is a severely disabling neurologic disorder with profound effects on productivity and health-related quality of life (HRQoL).<sup>7</sup> CM is currently defined as headache on  $\geq$ 15 days per month for >3 months, of which  $\geq$ 8 days have features of migraine headache.<sup>34</sup> Patients with CM suffer not only from the frequency, duration, and severity of headache and associated neurologic symptoms but also from multiple other dimensions that result in headache-related disability and impairment in headache-specific quality of life. Frequently, CM is associated with other pain conditions and comorbidities.<sup>27,36,50</sup> CM patients often seek care from pain specialists.<sup>7</sup> When examining treatment benefit of new agents for CM, it is valuable to examine all available scientific evidence of efficacy rather than focusing on a single efficacy dimension that may not fully reflect the clinical relevance of the outcome (ie, having a positive and meaningful impact on the patient's life).<sup>22</sup>

The use of all evidence to evaluate the clinical importance or meaningfulness of observed changes in outcome measures is particularly crucial in the study of chronic pain conditions, such as CM, in which 1) self-reported outcomes are inherently subjective; 2) improvements in physical and emotional functioning are often as important to patients as the actual diminution of pain; 3) results are frequently confounded by the use of rescue or concomitant pain treatments; and 4) placebo responses are often high.<sup>19,22</sup> Also, efficacy results usually are focused on evaluating a drug's mean benefit across a population. However, of equal or greater importance is the translation of mean changes in terms that better describe the effects for individual patients. Distribution of effects across the population is thus also very helpful to evaluate.<sup>20</sup>

A systematic method for interpreting the clinical importance (meaningfulness) of group differences in chronic pain studies has been recommended by the Initiative on Methods, Measurement, and Pain Assessment in Clinical Trials (IMMPACT), a consortium of participants from academia, government drug regulatory agencies, self-help groups, and the pharmaceutical industry who work to develop ways to improve the study of chronic pain conditions.<sup>22</sup> IMMPACT recommends that when evaluating the clinical meaningfulness of a treatment benefit, statistically significant group differences in a primary efficacy endpoint cannot be considered in isolation, as this may obscure meaningful individual patient improvements and other benefits and risks. Rather, the overall body of evidence with regard to outcomes must be considered to fully understand therapeutic benefit.<sup>22</sup> We have used an adaptation of the framework suggested by IMMPACT for evaluating clinical meaningfulness of treatment outcomes in chronic pain studies (Table 1).

Herein, we examine the clinical meaningfulness of both onabotulinumtoxinA and topiramate in the preventive treatment of CM in adults, focusing on the large Phase III REsearch Evaluating Migraine Prophylaxis Therapy (PREEMPT) clinical program<sup>4,19</sup> and the phase IV topiramate CM study.<sup>43</sup> The totality of the evidence as well as minimally important difference (MID; ie, the difference demonstrated to be clinically meaningful to patients) for any headache-symptom measures are discussed. Although our focus is on large studies, which requires that we make indirect comparisons across studies, we also review the limited data directly comparing onabotulinumtoxinA with topiramate in the preventive treatment of CM.

### Table 1. Adapted IMMPACT Framework for Evaluating Clinical Meaningfulness of Treatment Outcomes<sup>22</sup>

CATEGORY	Content
Efficacy	A. Headache symptom measures: statistical significance vs placebo
	B. Headache symptom measures: magnitude of improve- ment (individual patient benefit)
	C. Headache symptom measures: Responder analyses
	D. HRQOL measures
Safety	A. Safety and tolerability
Other	<ul> <li>A. Treatment effect size compared to available treatments</li> <li>B. Rapidity of onset and duration of treatment benefit</li> <li>C. Convenience and patient adherence to treatment</li> <li>D. Uniqueness of mechanism of action</li> <li>E. Drug-drug interactions</li> <li>F. Limitations of oral prophylaxis treatments</li> </ul>

Download English Version:

# https://daneshyari.com/en/article/2728822

Download Persian Version:

https://daneshyari.com/article/2728822

Daneshyari.com